Author's response to reviews

Title: The Casualty Chain Inventory: A new scale for Measuring Peritraumatic Responses: a cross-sectional study

Authors:

Laila Skogstad (laila.skogstad@uus.no)
Erlend Hem (erlend.hem@medisin.uio.no)
Leiv Sandvik (Leiv.Sandvik@uus.no)
Øivind Ekeberg (Oivind.Ekeberg@oslo-universitetssykehus.no)

Version: 6 Date: 12 May 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team (Emergency Medicine)

Thank you for accepting the manuscript. "The Casualty Chain Inventory: A new scale for Measuring Peritraumatic Responses: a cross-sectional study is in the proofing stage, and a new submission has been made. The changes have been made according to your request:

1) All tables are now cited in numerical order in the text, also Table 3
2) The tables are included in the manuscript
3) Highlighting is removed
4) Full citation details for reference 17 is provided
5) The manuscript is checked for typographical errors, also the abstract, and a few changes have been made

I hope the manuscript is correctly submitted. This Covering letter is sent separately, it was not possible to submit it. Please let me know if anything must be changed.

Best regards

Laila Skogstad
PhD candidat